Author's response to reviews

Title: Association study of the HLA-DRB1 locus reveals the first evidence for the association of HLA-DRB1*15 and DRB1*09 with leprosy and the impact of DRB1*09 on the onset of disease in Chinese population

Authors:

Furen Zhang (zhangfuren@hotmail.com)
Hong Liu (hongyue2519@hotmail.com)
Shumin Chen (chenshum@publicjn.sd.cn)
Changyuan Wang (wangchangyuan2005@yahoo.com)
Chuanfu Zhu (sdhla@yahoo.cn)
Hailin Zhang (zanghailin@yahoo.com.cn)
Tongsheng Chu (tongshengchu@163.com)
Dianchang Liu (liudianchang@yahoo.com.cn)
Xiaoxiao Yan (yanxiaoxiaoa1979@163.com)
Jianjun Liu (liuj3@gis.a-star.edu.sg)

Version: 2 Date: 19 February 2009

Author's response to reviews:

Dear editor:

We would like to submit our lab’s paper: Association study of the HLA-DRB1 locus reveals the first evidence for the association of HLA-DRB1*15 and DRB1*09 with leprosy and the impact of DRB1*09 on the onset of disease in Chinese population to your journal.

The data in the manuscript is original and the manuscript is not under consideration elsewhere. None of the manuscript contents have been previously published. All authors have read and approved all versions of the manuscript, its content, and its submission to BMC Medical Genetics. There is no Conflict of Interest. We will pay color publication fees if the paper is accepted for publication.

Correspondence to: Prof. Furen Zhang, Shandong Provincial Institute of Dermatovenereology, 57 Jiyan Lu, Jinan 250022, Shandong province, P.R. China
Tel: 86-0531-87298801 Fax: 86-0531-87984734

I am looking forward to hearing from you. Please don't hesitate to contact me if you have any questions.

Sincerely yours, Furen Zhang

Prof. Furen Zhang
Shandong Provincial Institute of Dermatovenereology. 57 Jiyan Lu, Jinan 250022, Shandong Provience, P.R. China